Author's response to reviews

Title: A pilot study of individualized comprehensive treatment for gastric cancer with para-aortic lymph node metastases

Authors:

Qi He (lheqinju@gmail.com)
Yang Li (li_yang82@icloud.com)
Chaogang Fan (241549864@qq.com)
Jian Xu (qiurong12@gmail.com)
Xixu Zhu (qiqi7310@yahoo.com.cn)
Guoli Li (liguoli15@163.com)

Version: 6 Date: 22 November 2015

Author's response to reviews: see over
Dear Ms Ducot, Dr. Bornschein and reviewers,

Thank you very much for your detailed review again and we revised the manuscript according to the comments.

1. We used some professional language editing service to improve language and grammar, including the typos mentioned by reviewer Dr. Duong.

2. We corrected the inconsistency of the protocol in the Methods and the Abstract.

3. We revised the incomplete sentences mentioned by reviewer Dr. Elsing. (Line85 and line222)

Thanks again for your detailed review and helpful suggestions.

Sincerely
Qi He
On behalf of Dr. Guoli Li